Author’s response to reviews

Title: Anemia and perioperative mortality in Non-cardiac Surgery patients: a secondary analysis based on a single-center retrospective study

Authors:

Xueying Luo (hjssums@126.com)
Feng Li (1052035175@qq.com)
Haofei Hu (huhaofei0319@126.com)
Baoer Liu (357028572@qq.com)
Sujing Zheng (498201584@qq.com)
Liping Yang (767198980@qq.com)
Rui Gao (1403646732@qq.com)
Ya Li (1137576813@qq.com)
Rao Xi (keybaby9@hotmail.com)
Jinsong He (hjssums1105@126.com)

Version: 2 Date: 06 Apr 2020

Author’s response to reviews:

Dear Editors and Reviewers:

Thank you for your letter and for the reviewers’ comments concerning our manuscript entitled “Anemia and perioperative mortality in Non-cardiac Surgery patients: a secondary analysis based on a single-center retrospective study”. (BANE-D-20-00084R1). Those comments are all valuable and very helpful for revising and improving our paper, as well as the important guiding significance to our researches. We have studied comments carefully and have made correction which we hope meet with approval. Revised portion are marked in red in the paper.

We tried our best to improve the manuscript and made some changes in the manuscript. These changes will not influence the content and framework of the paper.

We appreciate for Editors and Reviewers’ warm work earnestly, and hope that the correction will meet with approval.
Once again, thank you very much for your comments and suggestions.

Best wishes,

Jinsong He